Author’s response to reviews

Title: "Recruitment, recruitment, recruitment" - the need for more focus on retention: A qualitative study of five trials

Authors:
Anne Daykin (anne.daykin@bristol.ac.uk)
Clare Clement (c.clement@bristol.ac.uk)
Carrol Gamble (c.gamble@liverpool.ac.uk)
Anna Kearney (a.kearney@liverpool.ac.uk)
Jane Blazeby (j.m.blazeby@bristol.ac.uk)
Mike Clarke (m.clarke@qub.ac.uk)
Athene Lane (athene.lane@bristol.ac.uk)
Alison Shaw (ali.heawood@bristol.ac.uk)

Version: 1 Date: 19 Oct 2017

Author’s response to reviews:

Firstly we would like to thank the reviewers for their kind comments and recognition of the importance of our work.

All revisions have been highlighted in yellow within the manuscript.

Comment:

Page 2, Line 12

Revise sentence for clarity and readability: So, to understanding retention issues and strategies to address this is important."

Response:

This sentence has now been revised.

Comment:
Page 3, Line 34

Include "of" in the sentence - ...unaware of making ...

Response:

The word "of" has now been included.

Comment:

Include start/stop dates for data collection.

Response:

A sentence stating data collection period has now been added (page 5 under data collection).

Comment:

Was there any prior relationship between study researchers and trial's staff being interviewed?

Response:

Details of relationships have now been added to the data collection section (page 5).